Author’s response to reviews

Title: Incomplete ileus and hemafecia as the presenting features of multi-organ involved primary systemic AL amyloidosis: a rare case report

Authors:

Li Tian (fieldpower927@163.com)
Anliu Tang (13875893450@163.com)
Xian Zhang (1192463160@qq.com)
Zhen Mei (897449572@qq.com)
Fen Liu (liufencsu@163.com)
Jingbo Li (355704560@qq.com)
Xiaoyu Li (lixiayu@163.com)
Feiyan Ai (522688650@qq.com)
Xiaoyan Wang (wxy20011@163.com)
Shourong Shen (ssrxy3yy@163.com)

Version: 2 Date: 22 Nov 2016

Author’s response to reviews:

Dear Reviewers:

Thank you for reviewing our paper and your questions. We have seriously considered as well as responded last time. And for the revised part, we have marked in blue in the paper.

We considerably appreciate your consideration again. If you have any questions, please contact me without any hesitation.